Generated using the official AMS LATEX template v6.1 two-column layout. This work has been submitted for publication. Copyright in this work may be transferred without further notice, and this version may no longer be accessible.

Rapid topographic scatter of near-inertial waves generated by storms

ASHLEY J. BARNES,^{a,b} CALLUM J. SHAKESPEARE^{a,b}, ANDREW MCC. HOGG,^{a,b,c} and Navid C. Constantinou^{c,d}

^a Research School of Earth Sciences, Australian National University, Canberra, ACT, Australia
^b Australian Research Council Center of Excellence for Climate Extremes, Australia
^c Australian Research Council Center of Excellence for the Weather of the 21st Century, Australia
^d School of Geography, Earth and Atmospheric Sciences, University of Melbourne, Parkville, VIC, Australia

ABSTRACT: Internal waves propagate on the ocean's stratification, carrying energy and redistributing momentum through the ocean. When internal waves break, they contribute to diapycnal mixing in the ocean interior, but this breaking behaviour depends upon the scale of the waves. Low-mode internal waves have larger horizontal and vertical scales, and thus break less readily than higher-mode waves. The scattering of internal waves by topography is an important mechanism in transferring internal wave energy to smaller scales that are more conducive to wave breaking and mixing processes. In this study, we propose and investigate a mechanism in which storm-generated low-mode internal waves scatter at topography. We hypothesise that horizontally propagating internal wave modes generated by strong winds (i.e., due to a storm) can rapidly dephase; these dephased waves can then be scattered from topography, resulting in higher-mode upward-propagating waves within hours of the passage of a storm. We investigate this phenomenon in an idealised numerical model of a storm passing over a prominent ridge. Bottom-scattered near-inertial internal waves propagate away from the ridge rapidly in the wake of the storm. We perform several perturbation experiments varying the properties of the ocean, the winds and the topography. The bottom-scattered near-inertial waves in our domain. Although small in a globally averaged sense, we argue that the topographic scatter of storm-generated near-inertial waves could account for the unexplained near-inertial wave signals found in ocean observations and numerical studies.

SIGNIFICANCE STATEMENT: The ocean is stratified: denser waters are in the deep ocean and lighter waters closer to the surface. How and where waters of different densities mix is important as this contributes to global ocean circulation and how the ocean transports heat from the tropics to the poles. This ocean mixing is influenced greatly by the winds and tides that churn up the ocean near boundaries and generate internal waves, the breaking of which is thought to be the major source of turbulent mixing in the ocean interior. These waves propagate in the interior of the ocean rather than just at the surface, and are made possible by the ocean's stratification. These waves are very difficult to observe and so our understanding of their generation, propagation and breaking is an active field of research. Here, we demonstrate that internal waves generated by strong storms, like typhoons or cyclones, can rapidly scatter from nearby topography at the sea floor. Previous studies looked at waves generated by larger-scale winds rather than individual storms and suggested that the waves take weeks to reach the seafloor. Here, we show that storm-generated waves scatter at the seafloor within hours. Our study could help explain some of the bottomintensified wave activity that has been observed within hours of the passage of a storm. We find that the waves propagating upwards from topography have smaller spatial scales, and up to 10% the energy flux compared to the surface-generated waves that they scatter from. We argue

that this rapid wave-scattering mechanism is worthy of further study to better understand how they might fit into the ocean mixing picture.

1. Introduction

Internal waves, which manifest as mixed horizontal and vertical oscillations in a rotating stratified fluid, permeate the world's oceans. Internal waves are important to the fields of physical oceanography and climate science more broadly due to their significant contribution to ocean mixing and momentum transfer (Bell 1975; Alford et al. 2016; Sarkar and Scotti 2017; Thomas and Zhai 2022; Musgrave et al. 2022). Global estimates of the energy contained by the internal wave field are supported by sparse insitu observations and satellite altimetry (Zhao et al. 2016; Whalen et al. 2018). However, capturing the spatial variability of internal waves, and in turn their contribution to local episodic mixing, remains an ongoing challenge in oceanography, particularly in the deep ocean where observations are especially sparse. A further complication is that some types of internal wave can contribute to the local mixing at the generation site, while others can carry their energy great distances to enhance mixing in remote locations (Waterhouse et al. 2014).

Ocean models can provide insights where observations are lacking, and accurately capturing internal wave induced mixing in numerical models is a large area of study. Some approaches to this problem, such as Laurent and Garrett

Corresponding author: A. J. Barnes, ashley.barnes@anu.edu.au

(2002) and Lavergne et al. (2020), construct spatially varying maps of the wave field by parameterising different wave generation mechanisms. This method has shown promise in matching observations where they are available (Lavergne et al. 2020), but an unavoidable source of error in such approaches comes from processes that are omitted. By estimating the importance of this missing physics, we can increase our understanding of how well wave-induced mixing is represented in numerical models.

The bulk of internal waves begin their life-cycle at the boundary of the ocean - either at the surface, generated via winds (Thomas and Zhai 2022), or near the bottom as a result of flow over topography (Musgrave et al. 2022). In the first case, inertial oscillations at the surface are readily excited by the wind, which in turn generate nearinertial waves that can propagate downwards, beyond the mixed layer (D'Asaro 1985). These wind-driven, surfacegenerated near-inertial waves (NIWs) are responsible for between 0.3-1.1 TW of energy flux at the surface (Jiang et al. 2005; Rimac et al. 2013). The exact mechanism by which the near-inertial energy escapes the mixed layer depends significantly on the length scale of the wind forcing. Forcing by small, intense storm systems (Gill 1984) generates low-modes which propagate thousands of kilometers horizontally away from their generation site (Alford 2003). For larger scale winds, the near-inertial oscillations instead need to be downscaled by β -refraction or mesoscale interactions (Thomas and Zhai 2022) in order to escape the mixed layer. These waves propagate at very low vertical group speeds and thus shallow angles, with observational studies suggesting that vertical propagation takes weeks to bring such waves to the seafloor after generation (Kunze and Sanford 1984; Ma et al. 2022). Here, they may scatter, dissipate or reflect depending on the local topography (Igeta et al. 2009).

In the second case, topography generates internal waves directly. When deep ocean flows interact with topography, this can lead to vertical motions, and thereby the radiation of internal waves due to the restoring effect of the stratification. Examples of such interactions include the barotropic tides generating internal tides (Bell 1975), as well as lee waves excited by geostrophic currents (Nikurashin and Ferrari 2010), which are thought to have globally-integrated energy fluxes of 0.9-1.5 TW and 0.05-0.85 TW respectively (Nikurashin and Ferrari 2011; Egbert and Ray 2000; Waterhouse et al. 2014; Bennetts et al. 2024).

There is an indirect connection between wind-generated and topographically-generated internal waves. Wind forcing energises both long-lived and transient geostrophic currents (Wunsch 1998), which in turn shed energy at the ocean bottom via topographic lee waves (Nikurashin and Ferrari 2010). Nikurashin and Ferrari (2011) estimated that 20% of energy input to geostrophic flows through synoptic scale wind forcing was converted to lee wave generation. Resonant interactions of the lee waves and geostrophic flows (Nikurashin and Ferrari 2010) drive nearinertial energy and enhanced mixing near topographic features (Hu et al. 2020; Brearley et al. 2013; Liang and Thurnherr 2012). Given that the wind-driven geostrophic currents responsible for lee wave generation vary on timescales of months and years, this mechanism represents a persistent, steady source of internal wave generation.

In contrast, recent observations have found rapid generation of near-inertial waves at the seafloor within hours of a storm passing overhead. van Haren (2020) found evidence for enhanced mixing in the depths of the Mariana Trench, with a mooring detecting patterns in the stratification similar to that of breaking internal waves within hours of a category 4 typhoon. The rapid timescale and close proximity of the mixing to the storm suggest that neither of the two cases discussed above — that is, the interaction of lee waves with geostrophic flows nor the downwards propagation of near-inertial waves with low vertical group speeds excited by large scale winds — are able to account for the observations. As such, a new mechanism is needed.

In this work, we investigate the rapid scattering of nearinertial internal waves at the seafloor in the wake of a storm system. We hypothesise that the sharp horizontal gradients of storm systems allow for the generation of full-depth vertical modes, unlike other near-inertial waves generated by larger scale wind patterns as discussed in Kunze and Sanford (1984) and Ma et al. (2022). Other full-depth internal wave modes, like the baroclinic tide, are known to scatter at topography, which can redistribute their energy to smaller horizontal scales (Peter Müller and Naihuai Xu 1992; Laurent and Garrett 2002; Bühler and Holmes-Cerfon 2011; Kelly et al. 2013). Like these other full-depth internal waves, we demonstrate that the near-inertial waves generated by a storm can also scatter from local topography, with upwards propagating internal waves appearing almost immediately. This short timescale of the bottom response is of particular interest, as it makes this mechanism a contender for explaining the aforementioned enhanced bottom mixing coinciding with the typhoon.

The mechanism for this rapid topographic scattering is the dephasing of vertical modes. Gill (1984) studied the ocean response to a short-lived wind event, like the passage of a storm, that instantaneously accelerates the mixed layer and imprints onto multiple internal wave vertical modes. The dephasing of these vertical modes results in the rapid communication of the surface layer perturbation downwards below the thermocline. Given the full depth nature of the vertical modes generated by storm systems, we extend Gill's work to consider a rapid response at the sea floor.

In the presence of topography, we anticipate that the deep oscillatory motions driven by dephasing vertical modes will generate internal waves in a similar fashion to internal tides (Bell 1975). Like internal tides, this generation process can equally be thought of as the scattering of a large scale wave (barotropic tide, or surface-generated NIW) to a smaller scale wave (baroclinic tide, or topographically scattered NIW).

As discussed above, the scattering of near-inertial waves generated by large-scale winds is associated with a significant time lag due to them not exciting full-depth modes, and the low propagation angle of the waves as they escape the mixed layer. In contrast, the smaller horizontal scales of storm-generated near-inertial waves, which are the focus of this study, readily excite full-depth modes, and in turn we expect a rapid bottom response as these modes dephase. In the presence of sufficiently narrow and prominent topographic features, we further expect the waves to scatter to higher modes, facilitating energy downscaling which may ultimately lead to additional bottom mixing.

Given that storms convert large amounts of wind energy into surface layer oscillations and downward propagating near-inertial waves (Sanford et al. 2011; Whalen et al. 2018), the coincidence of a storm directly above topographic features would provide the most pronounced generation. The hypothesised topographic scatter of stormgenerated near-inertial waves have, to our knowledge, not been studied before and thus it is unknown whether they constitute a significant contribution to the ocean internal wave field and deep ocean mixing. To investigate this phenomenon, we run a numerical ocean model with idealised wind forcing and bathymetry. Under various topographic and wind forcing scenarios, we calculate the energy flux associated with these waves, and compare the magnitude of energy flux to that of surface-generated near inertial waves excited directly by the wind.

The next section reviews the theory of Gill (1984), and demonstrates the excitation of a bottom flow from the dephasing mechanism. In section 3, we describe the model setup, as well as the method we used for separating the waves of interest from the rest of the flow. Section 4 includes a qualitative demonstration of these bottom scattered waves, followed by a quantitative analysis of the wave energy for various perturbation experiments. Finally, we conclude in section 5 by arguing that this mechanism could be responsible for some of the unexplained observations of excess bottom mixing and vertical shear.

2. Theory

Gill (1984) modelled the ocean's immediate response to the passage of a storm as an initial velocity u_0 in the mixed layer, with zero velocity deeper in the water column. This initial condition can be thought of as an infinite set of orthonormal vertical modes each with a different vertical wavenumber m_n , which destructively interfere below the mixed layer. As these modes necessarily propagate at different speeds, they rapidly de-phase from one another, resulting in finite flows over the entire ocean depth. To demonstrate this mechanism, consider an idealised hydrostatic 2D model of the ocean with a mixed layer of depth h_m , total depth H, and constant stratification N, just after the passage of a storm with characteristic horizontal length scale L and associated wavenumber $k = 2\pi/L$. The initial surface response to the storm when t = 0 is given by an initial mixed-layer velocity with a maximum of u_0 , while the interior is at rest; that is,

$$u(x, z, t=0) = u_0 \cos(kx) \mathcal{H}(z+h_m), \tag{1}$$

where z = 0 corresponds to the ocean's surface and \mathcal{H} is the Heaviside function, equal to 0 when $z < -h_m$. Here, we follow the typical method of decomposition of the linearised Boussinesq equations into a set of vertical modes with boundary conditions du/dz = 0 at z = 0 and z = -H(Olbers et al. 2012). For an arbitrary stratification profile N(z), the eigenvalue problem cannot typically be solved analytically, but for the case of constant stratification N the eigenfunctions take the form $\phi_n(z) = \sqrt{2/H} \cos(n\pi z/H)$. The eigenvalues m_n correspond to the vertical wavenumbers $m_n = n\pi/H$ of the solution and are related to the horizontal wavenumber k (set by the spatial structure of the wind forcing) and frequency ω_n via the dispersion relation for internal waves

$$\omega_n^2 = f^2 + N^2 \frac{k^2}{m_n^2}.$$
 (2)

The baroclinic velocity can be expanded in vertical modes as:

$$u(x,z,t) = \sum_{n=1}^{\infty} a_n \phi_n(z) \cos(kx - \omega_n t), \qquad (3)$$

where the amplitudes a_n are obtained by projecting the initial velocity (1) onto the *n*th mode:

$$a_n = \int_{-H}^0 \phi_n(z) \, u_0 \mathcal{H}(z+h_m) \, \mathrm{d}z. \tag{4}$$

The barotropic part of the solution is the vertical mean of the initial condition. Note that this example represents the surface forcing not as a single isolated storm, but as an infinite wind band that repeats spatially at the characteristic length scale *L*. For the purposes of this idealised model, the horizontal structure is unimportant, so we simply consider the location x = 0 to investigate the interior response.

Each of the vertical modes propagates at a different frequency and phase speed according to the dispersion relation for internal waves (2) meaning that lower modes propagate faster in the horizontal. Initially, the linear combination of modes with amplitudes prescribed by (4) sum to give u_0 in the mixed layer and zero at depth. But as time progresses, the different propagation speeds mean that the



FIG. 1: a) Evolution of the initial surface layer velocity u(x, z, t = 0) for a reference case of N = 20|f| and a horizontal scale of $L = 2\pi/k = 100$ km. b) The bottom velocity response corresponding to the reference conditions, as well as for storms with decreased (20 km) and increased (1000 km) horizontal scale L. Velocities have been non-dimensionalized against u_0 , which in this case is 0.1 m s⁻¹ over a 200 m thick mixed layer h_m .

The manner in which the dephasing happens depends on the differences in frequency ω_n and in turn on stratification N and length scale L. Figure 1 shows the vertical structure of the solution at x = 0 for $h_m = 200$ m, H = 4000 m, $|f| = 10^{-4}$ s⁻¹ and N = 20|f| over 5 inertial periods. In this case, flow develops throughout the water column, including at the bottom, within 10% of an inertial period. Figure 1b shows that varying L affects both the amplitude and frequencies of resultant bottom flow, with the solution tending towards purely inertial oscillations as L increases. As propagating waves can be generated at topography with super-inertial oscillations, of particular interest are the stronger super-inertial motions that can be seen in the cases with smaller horizontal length scale.

The thickness of the surface layer h_m plays a role in redistributing the energy between vertical modes, but does not affect their fundamental dephasing behaviour. This simple model demonstrates that a wind-induced surface perturbation in velocity — of the correct scale — can almost instantaneously generate super-inertial oscillatory flow at the sea floor. Existing theory for internal tides can now be applied in evaluating the likely impact of the wind-induced oscillatory flow on secondary wave generation — or scatter — at the seafloor. The conversion rate C of energy from an oscillatory flow of amplitude U_0 to buoyancy frequency N, topography height h, and horizontal wavenumber k, based on linear wave theory, is given by (Jayne and St. Laurent 2001):

$$C = \frac{1}{4}\rho_0 U_0^2 h^2 Nk,$$
 (5)

where ρ_0 is the reference density. Although this mechanism differs from internal tides in that the oscillatory flow excited by winds is not of constant amplitude or frequency, (5) holds for our study to leading order if U_0 refers instead to the characteristic amplitude of the wind-induced velocity. Our numerical results are evaluated against this scaling relation in section 5.

3. Numerical model and analysis methodology

To study topographic near-inertial wave generation mechanism, we design an idealised numerical experiment as shown in Fig. 2. The Modular Ocean Model 6 (MOM6; Adcroft et al. (2019)) is run in an isopycnal configuration with 21 layers at 2 km horizontal resolution, total depth of 4000 m and horizontal extent of 4000 km by 4000 km. The zonal and meridional boundaries are re-entrant, but the domain is large enough that internal waves do not return to the region of study around the ridge during the simulations.

The top isopycnal layer mimics a mixed layer and has a thickness of 50 m in the default case; the second layer below mimics a thermocline and initially has a thickness of 150 m; the remaining 19 layers below have an initial thickness of 200 m. A Coriolis parameter of $f = -10^{-4} \text{ s}^{-1}$ corresponding to the Southern mid-latitudes is selected. The top layer has a density of 1018 kg/m³, followed by a large jump in density with an equivalent buoyancy frequency N = 200|f| to simulate a thermocline. The remaining 19 layers have a linearly increasing density to give a constant buoyancy frequency of 40|f|, an approximation of a typical buoyancy frequency observed beneath the thermocline (Reagan, James R. et al. 2021).

A challenge in studying wind-generated topographic near-inertial waves is that, even in a highly idealised domain, the waves exist in the presence of other flows in the inertial band. These include non-propagating inertial oscillations as well as near-inertial waves generated directly from the wind forcing that have not undergone any interactions with topography. To simplify the separation of wind-induced topographic near-inertial waves from these other near-inertial and inertial flows, the experiment was designed with a high degree of symmetry. The wind pulse is zonally symmetric, so that any zonally propagating waves are a result of interactions with the topography, and so this zonal component contains the waves of interest. For our control experiment, a zonally symmetric 1000 m tall Gaussian ridge with full width at half maximum (FWHM) W = 10 km, running north to south through the domain, is placed in the domain centre. These parameters for the cross-ridge and vertical scales are chosen to be consistent with a typical seamount (Wessel et al. 2010).

The model is initialised from rest and forced by a zonally-uniform 10-hour pulse of zonal wind stress given by

$$\tau(y,t) = \begin{cases} \tau_0 \, e^{-y^2/2\sigma^2} \sin^2\left(2\pi t/T_\tau\right) & 0 \le t \le T_\tau/2, \\ -\tau_0 \, e^{-y^2/2\sigma^2} \sin^2\left(2\pi t/T_\tau\right) & T_\tau/2 < t \le T_\tau, \\ 0 & T_\tau < t, \end{cases}$$
(6)

where τ_0 is the peak wind stress, $L_{\tau} = 2\sqrt{2\log(2)}\sigma$ is the FWHM of the wind stress, and T_{τ} the forcing duration, with default values of $T_{\tau} = 10$ hours and $L_{\tau} = 80$ km, consistent with observed tropical cyclone size distributions (Zhang and Chan 2023). The default wind stress value $\tau_0 = 10 \text{ Nm}^{-2}$ corresponds to the strength of a category 4 tropical cyclone with surface wind speed of 63 m s^{-1} . The wind pulse is focused at the meridional-centre of the domain, and the eastward and westward oscillation serves to force the surface layer without inducing a significant time-mean flow, providing a robust idealised model of the process of interest (i.e., high-frequency wind forcing by storms). A small westward time-mean flow, less than 10^{-6} ms⁻¹, is still present, but this is too small to meet the $|f| < kU_0$ condition required to support topographic lee waves.

In choosing to filter the waves by symmetry, a drawback is that our wind event lacks zonal gradients, restricting the divergence of the wind field to the meridional gradient. This formulation means that, while near-inertial waves are expected to be generated due to this sharp, unidirectional gradient, the magnitude of these waves may be significantly reduced compared to those generated by the winds of a true cyclone, which has both strong zonal and meridional gradients. However, as the scattered wave energy scales linearly with the incident waves to first order (see (5)), the comparison between the incident and scattered wave-fields remains as an important metric, even if the energy fluxes themselves are under-represented.

Employing the experiment's zonal and meridional symmetries to separate incident and scattered waves, we calculate the wave energy fluxes propagating outwards through a square centred on x_0 , y_0 with side lengths D. The surfacegenerated NIW energy flux, E_S , corresponds to the energy flux propagating meridionally through the northern and southern sides, whereas the energy flux of the NIWs resulting from interactions with topography, E_T , correspond to the wave energy fluxes through the eastern and western boundaries of the square. Energy fluxes are the product of the velocities u or v normal to the grid-cell face, and pressure *P*. The symmetry of the domain ensures that the energy of the topographically-generated near-inertial waves that are the focus of this study is captured within E_T .

To isolate the velocities u and pressures P required to calculate E_T , we subtract reference values from 2000 km west of the ridge, such that $u' = u - u_{far}$ and likewise for P. Multiplying the perturbation velocity and pressure, and summing over the opposing sides of the square yields the zonal $E_{i,T}$ and meridional $E_{i,S}$ energy fluxes at the *i*th layer:

$$E_{i,T}(y,t) = \left[P'_{i}u'_{i}\right]_{x=\frac{D}{2}} - \left[P'_{i}u'_{i}\right]_{x=-\frac{D}{2}},\tag{7}$$

$$E_{i,S}(x,t) = [P_i v_i]_{y=\frac{D}{2}} - [P_i v_i]_{y=-\frac{D}{2}}.$$
 (8)

These energy fluxes are then integrated in space and time to determine the total energy leaving the square meridionally and zonally. The quantities of interest are the vertical energy flux from the storm directly into the surface-generated NIWs, and the vertical energy flux emanating upwards from the topography. Rather than directly calculating these surface and topographic vertical energy fluxes, which is challenging in an isopycnal model with layers intersecting with the topography, we instead assume that these energy fluxes are equivalent to the corresponding horizontal energy fluxes. This assumption means that the calculated energy fluxes will be underestimated by any dissipation or breaking that occurs during the horizontal propagation, but in this idealised model with no background flow and otherwise smooth topography, dissipative mechanisms are minimal. To allow time for the signal to propagate to the edges of the square, the analysis period extends 90 hours beyond the end of the storm but is still normalised by the 10 hour storm duration.

With this assumption, we then define our average vertical energy flux quantities $\overline{E_T}$ and $\overline{E_S}$:

$$\overline{E_T} = \frac{1}{T_{\tau}DW} \sum_{i=1}^{N} h_i \int_{-\frac{D}{2}}^{\frac{D}{2}} \int_{0}^{T} E_{i,T} \, \mathrm{d}t \, \mathrm{d}y, \qquad (9)$$

$$\overline{E_S} = \frac{1}{T_\tau D L_\tau} \sum_{i=1}^N h_i \int_{-\frac{D}{2}}^{\frac{D}{2}} \int_0^T E_{i,S} \, \mathrm{d}t \, \mathrm{d}x, \qquad (10)$$

where h_i is the *i*th layer thickness, *T* is the total simulation duration, and T_{τ} the duration of the storm. The experiment is run for 90 hours after the end of the storm, giving a total duration of $T = T_{\tau} + 90$ hrs, equivalent to 100 hours in the default case. Rather than divide by depth *H* to get the average energy flux through the sides of the square, factor 1/W in (9) converts the depth-integrated (summed over each isopycnal layer) horizontal energy flux to an equivalent vertical flux over the topography, providing a quantity in units of power per unit area of the topography (usually expressed in W/m²). Likewise, the factor $1/L_{\tau}$ converts



FIG. 2: A schematic of the model configuration, including the isopycnal displacements associated with both the surface and bottom generated near-inertial wave responses. Note that the vertical scales of the waves have been exaggerated, and that two layers are shown separately and in different colours for illustrative purposes, having been filtered using the zonal symmetry of the surface near-inertial wave response. The zonal and meridional centre lines as referred to in the text are denoted by x_0 and y_0 respectively.

the energy flux in meridional waves into a downward flux per unit area of the storm. These quantities are referred to as the surface and topographic energy fluxes throughout the study.

The choice of side length D for our square of energy flux integration needed to be close enough to the ridge that waves could travel a distance D/2 well before any waves would reflect back from the sides of the domain. Conversely, D must be larger than all of the ridge widths tested. Through testing several values of D, we found that 500 km was large enough to avoid any large amplitude, non-linear effects near the ridge, yet remaining within reasonable computational constraints concerning the required domain size.

4. Results

We first provide a qualitative demonstration of the proposed rapid scatter of near-inertial waves at topography. Fig. 3 shows the meridional velocities and isopycnal height anomalies associated with the meridionally propagating near-inertial waves generated directly at the surface. Internal waves are immediately visible through the full water column, similar to the analytic theory shown in Fig. 1. As time progresses, we see that the mode 1 waves (characterised by the single zero crossing) dominant in panel (a) give way to third and fourth vertical modes in panels (c) and (d). This behaviour is consistent with the modal dephasing theory in that the faster, lower modes appear first, which propagate more rapidly from the centre of the storm. By inspecting the phase shifts of individual vertical modes and the beam-like structures, it is clear that the frequency is near-inertial. Importantly to the generation of waves from topography, Fig. 3 shows significant motions at the seafloor immediately after the storm event.

These oscillatory motions are more clearly visible in Fig. 4a, which shows bottom velocities directly beneath the wind forcing similar to the 2D analytic theory in Fig. 1b. Here, the model shows an immediate bottom response with a clear super-inertial signal in the bottom flow, which enables the generation of propagating internal waves in the presence of appropriate topography. A Hovmöller diagram in Fig. 4b shows a signal in the resulting zonal velocity propagating away from the ridge, including several superimposed vertical modes characterised by their different group speeds, fanning out from the topography.

To visualise the bottom scattered waves, the velocity and interfacial height anomalies along two zonal transects are plotted in Fig. 5. Here, as in (7), the anomalies are calculated with respect to their far-field values to remove the effects of non-topographically-generated waves. In all panels, clear beams can be seen propagating upwards and



FIG. 3: Meridional transect showing meridional velocity and exaggerated isopycnal displacements associated with windgenerated NIWs. Isopycnal displacements are not to scale; they are exaggerated for visualization purposes.



FIG. 4: Panel a) Bottom zonal velocity 1000 km West of ridge, b) Hovmöller diagram of zonal velocity anomalies 5 layers from the bottom in the vicinity of the ridge with the approximate group speeds for vertical modes 1, 2, 3, 4, 6, and 9. Both panels show velocities 200 km north of the storm centre, as in the bottom row of Fig. 5.

outwards from the top of the ridge. The two rows, corresponding to locations 150 and 200 km northwards from the centre of the wind forcing, show meridional variations to the waves structure and phase. This is due to the time taken for the dephasing signal to propagate outwards from the storm, as well as the meridional structure of the storm itself.

To assess the role of these bottom scattered waves in redistributing energy to higher vertical modes and smaller scales, we take the power spectra of the velocities associated with the meridional and zonal waves in Fig. 6. The



FIG. 5: Zonal transect showing zonal velocity (color) and exaggerated isopycnal displacements (lines) associated with wind-generated topographic NIWs from our reference numerical simulation at four different time snapshots and two locations north of the storm.

prominent vertical modes of the surface-generated, meridional waves visible in Fig. 6a are consistent with the dispersion relation (2), and show a dominant first vertical mode, with energy also spread through modes 2-5. The spectrum of the scattered waves, shown in Fig. 6b, shows bands of power close to harmonics of |f|, with local maxima where these bands intersect vertical modes. Compared to the incident meridionally propagating waves, the scattered, zonally propagating waves have a very weak mode 1, and the circled regions in panel b show significant energy present in modes 6-10. Thus energy is distributed further towards higher modes in the scattered zonal waves than the incident meridionally propagating waves. This difference in the modal energy distribution shows that the bottom scatter of NIWs generated at the surface by storms can play a role in downscaling the energy to higher vertical modes, and in turn smaller horizontal scales.

Figure 6a shows that the incident meridional waves have most energy close to |f|, but with significant energy in the first harmonic. This enhanced first harmonic can be explained by the wind time scale of the storm – at 10 hours, the dominant frequency is close to 2|f|, a choice to ensure that most of the wind energy spectrum is super-inertial, in order to generate the most wave activity. In contrast to Fig. 6a, in Fig. 6b, we see that the first harmonic is more energetic than the near-inertial band, suggesting that this dephasing mechanism for bottom scatter is more efficient for 2|f|.

This observation is consistent with the strength of the superinertial component of the bottom oscillations compared to the purely inertial in both Fig. 4a and Fig. 1. Here, higher harmonics of |f| are identifiable as the addi-



FIG. 6: Depth averaged power spectra with the first 10 analytic vertical near-inertial modes overlaid in dashed grey of: a) the meridional velocity associated with the incident surface-generated near inertial waves b) the zonal velocity associated with the topographically scattered waves. Both spectra are taken over the first 10 inertial periods of the experiment, and at the same spatial positions as the corresponding waves shown in figures 3 and 5. The ovals in panel b highlights additional energy in modes 6-10 not present in panel a.

tional, consistently spaced zero crossings between inertial periods. These higher harmonics in Fig. 4a are modulated by by a near-inertial signal of comparable amplitude, much like the two smaller wind forcing scale cases in Fig. 1. The excitation of waves at 2|f| is likely due to higher harmonics dephasing more readily than lower: since the spatial scales imposed by the topography remain constant between the |f| and 2|f| waves, their group speeds remain the same, resulting in the faster dephasing of the 2|f| harmonic due to

it having more cycles per unit of distance. This explains the prominence of the 2|f| harmonic in this case of a transient storm system generating a rapid bottom response.

The waves shown in Figs. 5 and 4 have characteristics consistent with the proposed mechanism, and Fig. 6 demonstrates their potential for the downscaling of wave energy. Thus, these results serve as a qualitative demonstration of wind-induced topographically generated nearinertial waves. Having qualitatively demonstrated the rapid scatter of storm-generated NIWs at topography, we now quantify the wave energy, in order to assess in what circumstances, if any, these waves might pose a significant source of bottom intensified wave activity.

a. Wave Energy Flux

Having established the mechanism for the rapid topographic scattering of storm generated near-inertial waves in our configuration, we now proceed to calculate the corresponding energy fluxes of the incident and scattered waves. From (7), the time averaged energy flux in our reference configuration (Figs. 3, 5, and 4 above) associated with the wind-generated topographic NIWs is 0.13 mW m^{-2} , compared to 1.9 mW m⁻² for the wind-generated surface NIWs. As discussed in section 3, these energy flux values are an underestimate of what might be expected in the real ocean due to the simplified, zonally symmetric wind forcing that enables our clean wave separation. Thus, a more important metric is in comparing the relative sizes of the incident and scattered wave energy fluxes. With the control set of parameters - namely with a ridge 1000m high and 10km wide, and 63 ms⁻¹ winds lasting for 10 hours – the zonally propagating, scattered waves have a time-averaged energy flux 7% the size of that of meridionally propagating surface-generated NIWs.

b. Perturbation experiments

We run experiments varying both the properties of the storm and ridge to assess which parameters the energy flux is most sensitive to and determine the conditions under which the topographically-generated waves are most significant.

When altering the duration of the wind forcing pulse (Fig. 7a) there is a peak in both topographic and surface energy fluxes when the forcing duration is less than the inertial period, i.e., $2\pi/|f| \approx 17.5$ hours. This result is consistent with expectations for surface-driven near-inertial waves, as internal wave generation is only supported by super-inertial forcing frequencies. Forcing durations right at the inertial frequency would overlap significantly with the sub-inertial band, whereas a more rapid surface forcing would contain less energy at the highly resonant inertial period. The offset peaks in Fig. 7a) are likely due to the exciting of the 2|f| harmonic in the scattered waves, shown in Fig. 6b. This is due to the significant 2|f| component

of the bottom velocity (see Fig. 4a) which in turn is caused by the faster dephasing of the higher harmonic waves than the near-inertial.

The strength of the wind forcing pulse was altered in Fig. 7b, ranging from winds expected during a mild storm up to a category 5 tropical cyclone. Here, both the topographic and surface wave energy fluxes increase quadratically (with correlation coefficient 0.95). This quadratic relationship matches the wind work applied at the surface, which scales quadratically with the surface wind speed.

The second series of tests conducted relate to the shape of the topography (Fig. 7c,d). As expected from equation (5), the energy flux scales quadratically (with correlation coefficient 0.95) with ridge height up to 750 m, or just shy of a quarter of the water column depth. The energy flux then appears to saturate, likely due to the larger topography blocking the cross-ridge flow, and thereby damping wave generation (Winters and Armi 2014).

Finally, Fig. 7d shows the sensitivity of topographic energy flux when varying the ridge width. Here, the scaled energy flux, as in (7), is shown alongside the product of the energy flux and width. This plot is designed to show the way that the total wave energy changes, as well as energy flux, which scales inversely to ridge width. Here, we see that the energy flux per unit area (Fig. 7d, left axis) is greatest for the narrowest ridge, but that the total wave energy per metre meridionally (Fig. 7d, right axis) peaks for a 30 km wide ridge. This effect is due to the maximum topographic slope becoming shallower than the angle of the wave beam, suppressing the downward propagating part of the wave-field visible in Fig. 5 (result not shown). In (5), krefers to the dominant wavenumber of the ridge, which in our case is $k \sim 1/W$. This is consistent with the our results in Fig. 7, whereby the energy flux per unit area is greater for steeper and narrower topography.

5. Discussion and conclusions

We demonstrate a novel mechanism for the generation of wind-driven near-inertial waves at topography within hours of the passage of a storm. These waves are generated by the rapid dephasing of horizontally propagating full-depth wave modes induced by a transient wind event, and interaction of the resulting bottom flow with topography to radiate waves. This generation mechanism is showcased using an isopycnal model forced by a wind pulse that mimics a storm. The idealised model setup is carefully constructed to enable a clean separation of the scattered waves from surface-generated near-inertial waves and to eliminate any significant time-mean flow that could generate lee waves.

The perturbation experiments shown in Fig. 7 reveal that the variation of wave energy fluxes in our experiment is consistent with the existing internal wave theory. With our control experiment, namely a 1000 m-high, 10 km wide ridge in a 4000 m deep ocean, with an 80 km wide



FIG. 7: The energy flux of the wind-driven topographic near-inertial (zonal) waves when varying a) forcing duration; b) surface forcing wind speed; c) ridge height; d) ridge width. The default values are shown with an \times . The energy flux for the wind-driven surface-generated near-inertial (meridional) waves is shown in panels a) and b) only, as it is unaffected by varying the shape of the topography.

(FWHM) category 4 tropical cyclone overhead, the topographically scattered near-inertial waves have energy fluxes of 0.13 mW/m^2 , that is, 7% the energy flux in surfacegenerated NIWs in the same experiment. Given that our simplified storm under-represents surface NIW generation, we focus on the comparison between the surface-generated and topographically scattered waves, and consider these energy flux estimates as a lower bound.

With the most narrow topography that we test, the maximum energy flux attained is 0.39 mW/m^2 , or 18% of the surface-generated NIWs. However, as such a prominent ridge or seamount would be uncommon, we consider as a more typical range, the energy flux ratios from the strength and duration perturbation experiments (Fig. 7a,b). From these two experiments, the ratios of the topographically generated to the directly surface-generated NIW energy fluxes range from 5% at the longest duration and strongest storm, to 10% at the shortest and weakest storms. This range is consistent with the 6% of surface NIW energy estimated to reach the seafloor in the work by Jouanno et al. (2016), except that with our proposed mechanism this energy is communicated rapidly (within 100 hours of the storm) to the bottom.

Given that the mechanism described in this study relies on specific and short-lived circumstances, namely a storm event near prominent topography, it is not expected to be a major contributor to the internal wave field in a globally averaged sense. Instead, the mechanism could be important for explaining intermittent locally enhanced internal wave generation at the seafloor and deep-ocean mixing. Recent studies have identified evidence for unexplained enhancement of internal wave activity at the seafloor. van Haren (2020) finds anomalously large bottom near-inertial oscillations, mixing and deep overturning at 3000 m depth 1000 km away from a category 4 tropical cyclone. This bottom signal's first detection and intensification happened simultaneously with the approach of the cyclone, pointing to a rapid (on the order of hours or a few days) communication of the storm's energy to the deep ocean. Similarly, in another observational study, Morozov and Velarde (2008) show a rapid bottom response coinciding with the arrival of two cyclones, in addition to the downward propagating wave-packet which takes two weeks to reach the seafloor (see Figs. 11 and 12 by Morozov and Velarde (2008)). A modelling study of storm-induced flows shows enhanced near-inertial energies at the seafloor within 5 days of the passage of a storm, several days ahead of the downward propagating wave packets (see Fig. 9 by Jouanno et al. (2016)). Jouanno et al. (2016) note that while this bottom NIW generation is linked to the storm, the associated energies are two orders of magnitude smaller than the surfacegenerated NIWs and were thus not investigated further. However, our perturbation experiments suggest this ratio of surface to bottom generated NIW energy flux could be an order of magnitude larger than was found by Jouanno et al. (2016), warranting further investigation. We argue that the mechanism proposed in the study could at least partly account for these findings in the existing literature, but further study is needed to clarify the extent to which this may be the case.

Here, we studied the generation of wind-driven topographic near-inertial waves in a highly idealised domain to cleanly separate them from other flow features. An important next step is to identify the waves in more realistic simulations. However, delineating topographic nearinertial waves from other flow factors is difficult due to their generation by the dephasing of the vertical modes of the surface response, which itself includes NIWs of similar frequencies and spatial scales. Our experiments were carefully designed to separate the wind-driven topographic near-inertial waves from other flow features as cleanly as possible. Consequently, this caused the underestimation of the near-inertial wave energy imparted by the storm and limited the scope of our study to simple dynamics, topographic features, and surface forcing profiles. To investigate topographic near-inertial waves in a more realistic numerical simulation or with observations, these intricacies of disentangling them from other flow features will need to be handled carefully. One option could be to exploit the difference in vertical propagation between the topographic and wind-generated near-inertial waves. If a filter were used to separate the upwards and downwards propagating signals, perhaps following the spectral approach employed by Olbers and Eden (2017) or Waterhouse et al. (2022), one could identify areas of increased vertical energy flux beneath storm systems. However, one would still need to remove the background internal wave fields, among other oscillatory signals that could obscure

the waves of interest. To obtain a stronger signal, a statistical approach could be employed, e.g., averaging over many storm events in several locations above regions of varying topographic prominence. At each location, measurements of the bottom mixing could be taken both in the absence and presence of an overhead storm. Future studies could employ this approach on high-resolution model output or observations.

As shown in Fig. 6, the scatter of storm-generated NIWs at topography can redistribute energy towards higher vertical modes, and smaller horizontal scales. The higher vertical modes visible in Fig. 6b suggests that these scattered waves may be especially important for abyssal mixing due to their role in energy downscaling. However, our study was designed to maximise both wave generation and propagation so that the scattered wave-field was easier to isolate and study and therefore mechanisms for wave breaking and dissipation were intentionally either removed or suppressed. Hence we made no attempt to estimate abyssal mixing generated by this mechanism.

In addition to the spatial downscaling, the scattered waves tend to higher frequencies, particularly the prominent first harmonic of |f|. The reason for this is that the power spectrum of the bottom velocity from the dephasing of the incident waves in Fig. 4 contains a 20% stronger 2|f| peak than |f| (figure not shown), likely because the higher harmonics dephase more quickly and so are more prominent in the post-storm bottom velocity. As higher frequency internal waves exhibit less vertical shear, they are less prone to breaking, so this could somewhat offset any increase to mixing attributed to the spatial downscaling. Higher harmonics in the downscaling of near-inertial waves by eddies was observed in Danioux et al. (2011). In this study, authors found that the 2|f| harmonic dominated |f| at depth, stating that the mechanism explaining this shift in frequency in depth remains unresolved. If the role of eddies in downscaling these near-inertial waves also involves the dephasing of vertical modes, then the results of our study could help to explain these findings.

The results of our idealised study suggest that the scatter of storm-generated NIWs at topography may contribute to energy downscaling and in turn deep ocean mixing, but the exact extent and location of enhanced bottom mixing due to our proposed mechanism remains an open question. It remains possible that in more realistic conditions the waves would immediately break, become trapped, or otherwise dissipate close to the generation site, which was intentionally avoided in our study to better measure the energy fluxes. Further study is therefore needed to understand how these waves would behave in the real ocean. If windforced topographic near-inertial waves are responsible for energy downscaling and intensified near-bottom mixing, then one could assess their impact indirectly by measuring the wavenumber spectra and bottom mixing rates at other intersection points of topographic features and storm systems. This could be done either with existing moorings, or by analysing model outputs of high resolution ocean models with more realistic forcing and bathymetry. The advantage of using high-resolution ocean models is that the effects of tides could easily be removed, so enhanced internal wave generation or breaking at the bottom in the presence of storms could more easily be teased out. It would also be worthwhile to apply these analysis techniques to the data of the previously mentioned studies (van Haren 2020; Morozov and Velarde 2008; Jouanno et al. 2016) which found evidence for bottom NIW activity immediately after a storm, to investigate whether this mechanism is indeed consistent with these observations.

In conclusion, we have studied a novel internal-wave generation mechanism whereby the rapid dephasing of a wind-driven internal waves modes results in bottom flow over topographic features. The energy flux associated with these wind-driven topographic near-inertial waves reached 0.2 mWm^{-2} , a value 10% the size of surface-generated NIWs in the same domain. By separating the topographic near-inertial waves from other dynamical features in our idealised experiments, we found that wave energy scales in a way consistent with existing internal waves theory. Further study of topographic near-inertial waves in ocean observations and realistic models would help us clarify how these waves might fit into the broader internal wave spectrum, and our picture of abyssal mixing.

Acknowledgments. Computational resources were provided by the National Computational Infrastructure at the Australian National University, which is supported by the Commonwealth Government of Australia. A.J.B. acknowledges their PhD stipend provided by the Australian Research Training Program. We acknowledge funding from the Australian Research Council under DECRA Fellowship DE210100749 (N.C.C.), the Centre of Excellence for Climate Extremes CE170100023 (A.J.B., A.M.H., and C.J.S.), and the Centre of Excellence for the Weather of the 21st Century CE230100012 (A.M.H. and N.C.C.). We would like to thank the two anonymous reviewers and the editor for their constructive comments that greatly improved the manuscript.

Data availability statement. The model configuration setup, post processing, analysis and figure rendering code is publicly available at github.com/ ashjbarnes/topographic-NIWs, and replicated in a Zenodo repository at https://doi.org/10.5281/ zenodo.15493445. The MOM6 source code can be found at github.com/mom-ocean/MOM6.

References

Adcroft, A., and Coauthors, 2019: The GFDL Global Ocean and Sea Ice Model OM4.0: Model Description and Simulation Features. *Journal of Advances in Modeling Earth Systems*, **11** (10), 3167–3211, https://doi.org/10.1029/2019MS001726.

- Alford, M. H., 2003: Redistribution of energy available for ocean mixing by long-range propagation of internal waves. *Nature*, **423** (6936), 159–162, https://doi.org/10.1038/nature01628.
- Alford, M. H., J. A. MacKinnon, H. L. Simmons, and J. D. Nash, 2016: Near-Inertial Internal Gravity Waves in the Ocean. *Annual Review of Marine Science*, 8 (1), 95–123, https://doi.org/10.1146/ annurev-marine-010814-015746.
- Bell, T. H., 1975: Topographically generated internal waves in the open ocean. Journal of Geophysical Research (1896-1977), 80 (3), 320– 327, https://doi.org/10.1029/JC080i003p00320.
- Bennetts, L. G., and Coauthors, 2024: Closing the Loops on Southern Ocean Dynamics: From the Circumpolar Current to Ice Shelves and From Bottom Mixing to Surface Waves. *Reviews of Geophysics*, 62 (3), e2022RG000 781, https://doi.org/10.1029/2022RG000781.
- Brearley, J. A., K. L. Sheen, A. C. N. Garabato, D. A. Smeed, and S. Waterman, 2013: Eddy-Induced Modulation of Turbulent Dissipation over Rough Topography in the Southern Ocean. *Journal* of *Physical Oceanography*, **43** (11), 2288–2308, https://doi.org/ 10.1175/JPO-D-12-0222.1.
- Bühler, O., and M. Holmes-Cerfon, 2011: Decay of an internal tide due to random topography in the ocean. *Journal of Fluid Mechanics*, 678, 271–293, https://doi.org/10.1017/jfm.2011.115.
- Danioux, E., P. Klein, M. W. Hecht, N. Komori, G. Roullet, and S. L. Gentil, 2011: Emergence of Wind-Driven Near-Inertial Waves in the Deep Ocean Triggered by Small-Scale Eddy Vorticity Structures. *Journal of Physical Oceanography*, **41** (7), 1297–1307, https://doi.org/10.1175/2011JPO4537.1.
- D'Asaro, E. A., 1985: The Energy Flux from the Wind to Near-Inertial Motions in the Surface Mixed Layer. *Journal of Physical Oceanography*, **15 (8)**, 1043–1059, https://doi.org/10.1175/1520-0485(1985) 015(1043:TEFFTW)2.0.CO;2.
- Egbert, G. D., and R. D. Ray, 2000: Significant dissipation of tidal energy in the deep ocean inferred from satellite altimeter data. *Nature*, 405 (6788), 775–778, https://doi.org/10.1038/35015531.
- Gill, A. E., 1984: On the Behavior of Internal Waves in the Wakes of Storms. *Journal of Physical Oceanography*, **14** (7), 1129–1151, https://doi.org/10.1175/1520-0485(1984) 014(1129:OTBOIW)2.0.CO;2.
- Hu, S., and Coauthors, 2020: Dynamic features of near-inertial oscillations in the Northwestern Pacific derived from mooring observations from 2015 to 2018. *Journal of Oceanology and Limnology*, **38** (4), 1092–1107, https://doi.org/10.1007/s00343-020-9332-1.
- Igeta, Y., Y. Kumaki, Y. Kitade, T. Senjyu, H. Yamada, T. Watanabe, O. Katoh, and M. Matsuyama, 2009: Scattering of near-inertial internal waves along the Japanese coast of the Japan Sea. *Journal of Geophysical Research: Oceans*, **114** (C10), https://doi.org/ 10.1029/2009JC005305.
- Jayne, S. R., and L. C. St. Laurent, 2001: Parameterizing tidal dissipation over rough topography. *Geophysical Research Letters*, 28 (5), 811– 814, https://doi.org/10.1029/2000GL012044.
- Jiang, J., Y. Lu, and W. Perrie, 2005: Estimating the energy flux from the wind to ocean inertial motions: The sensitivity to surface wind fields. *Geophysical Research Letters*, **32** (15), https://doi.org/ 10.1029/2005GL023289.

- Jouanno, J., X. Capet, G. Madec, G. Roullet, and P. Klein, 2016: Dissipation of the energy imparted by mid-latitude storms in the Southern Ocean. *Ocean Science*, **12** (3), 743–769, https://doi.org/ 10.5194/os-12-743-2016.
- Kelly, S. M., N. L. Jones, and J. D. Nash, 2013: A Coupled Model for Laplace's Tidal Equations in a Fluid with One Horizontal Dimension and Variable Depth. *Journal of Physical Oceanography*, **43** (8), 1780–1797, https://doi.org/10.1175/JPO-D-12-0147.1.
- Kunze, E., and T. B. Sanford, 1984: Observations of Near-Inertial Waves in a Front. *Journal of Physical Oceanography*, **14** (**3**), 566– 581, https://doi.org/10.1175/1520-0485(1984)014(0566:OONIWI) 2.0.CO;2.
- Laurent, L. S., and C. Garrett, 2002: The Role of Internal Tides in Mixing the Deep Ocean. JOURNAL OF PHYSICAL OCEANOGRAPHY, 32, 18.
- Lavergne, C., and Coauthors, 2020: A Parameterization of Local and Remote Tidal Mixing. *Journal of Advances in Modeling Earth Systems*, **12** (5), https://doi.org/10.1029/2020MS002065.
- Liang, X., and A. M. Thurnherr, 2012: Eddy-Modulated Internal Waves and Mixing on a Midocean Ridge. *Journal of Physical Oceanography*, 42 (7), 1242–1248, https://doi.org/10.1175/JPO-D-11-0126.1.
- Ma, Y., D. Wang, Y. Shu, J. Chen, Y. He, and Q. Xie, 2022: Bottom-Reached Near-Inertial Waves Induced by the Tropical Cyclones, Conson and Mindulle, in the South China Sea. *Journal of Geophysical Research: Oceans*, **127** (6), e2021JC018162, https://doi.org/ 10.1029/2021JC018162.
- Morozov, E. G., and M. G. Velarde, 2008: Inertial oscillations as deep ocean response to hurricanes. *Journal of Oceanography*, **64** (4), 495– 509, https://doi.org/10.1007/s10872-008-0042-0.
- Musgrave, R., F. Pollmann, S. Kelly, and M. Nikurashin, 2022: Chapter 6 - The lifecycle of topographically-generated internal waves. *Ocean Mixing*, M. Meredith, and A. Naveira Garabato, Eds., Elsevier, 117–144, https://doi.org/10.1016/B978-0-12-821512-8. 00013-X, URL https://www.sciencedirect.com/science/article/pii/ B978012821512800013X.
- Nikurashin, M., and R. Ferrari, 2010: Radiation and Dissipation of Internal Waves Generated by Geostrophic Motions Impinging on Small-Scale Topography: Theory. *Journal of Physical Oceanography*, 40 (5), 1055–1074, https://doi.org/10.1175/2009JPO4199.1.
- Nikurashin, M., and R. Ferrari, 2011: Global energy conversion rate from geostrophic flows into internal lee waves in the deep ocean. *Geophysical Research Letters*, **38** (8), https://doi.org/10.1029/ 2011GL046576.
- Olbers, D., and C. Eden, 2017: A Closure for Internal Wave–Mean Flow Interaction. Part I: Energy Conversion. *Journal of Physical Oceanography*, **47** (6), 1389–1401, https://doi.org/10.1175/JPO-D-16-0054. 1.
- Olbers, D., J. Willebrand, and C. Eden, 2012: Ocean Dynamics. Springer Berlin Heidelberg, Berlin, Heidelberg, https://doi.org/10. 1007/978-3-642-23450-7, URL http://link.springer.com/10.1007/ 978-3-642-23450-7.
- Peter Müller, and Naihuai Xu, 1992: Scattering of Oceanic Internal Gravity Waves off Random Bottom Topography. *Journal of Physical Oceanography*, 22 (5), 474–488.

- Reagan, James R., and Coauthors, 2021: World Ocean Atlas. URL https://www.ncei.noaa.gov/products/world-ocean-atlas.
- Rimac, A., J.-S. von Storch, C. Eden, and H. Haak, 2013: The influence of high-resolution wind stress field on the power input to near-inertial motions in the ocean. *Geophysical Research Letters*, **40** (18), 4882– 4886, https://doi.org/10.1002/grl.50929.
- Sanford, T. B., J. F. Price, and J. B. Girton, 2011: Upper-Ocean Response to Hurricane Frances (2004) Observed by Profiling EM-APEX Floats. *Journal of Physical Oceanography*, **41** (6), 1041–1056, https://doi.org/10.1175/2010JPO4313.1.
- Sarkar, S., and A. Scotti, 2017: From Topographic Internal Gravity Waves to Turbulence. *Annual Review of Fluid Mechanics*, 49 (1), 195–220, https://doi.org/10.1146/annurev-fluid-010816-060013.
- Thomas, L. N., and X. Zhai, 2022: Chapter 5 The lifecycle of surfacegenerated near-inertial waves. *Ocean Mixing*, M. Meredith, and A. Naveira Garabato, Eds., Elsevier, 95–115, https://doi.org/10.1016/ B978-0-12-821512-8.00012-8, URL https://www.sciencedirect. com/science/article/pii/B9780128215128000128.
- van Haren, H., 2020: Challenger Deep internal wave turbulence events. Deep Sea Research Part I: Oceanographic Research Papers, 165, 103 400, https://doi.org/10.1016/j.dsr.2020.103400.
- Waterhouse, A. F., and Coauthors, 2014: Global Patterns of Diapycnal Mixing from Measurements of the Turbulent Dissipation Rate. *Journal of Physical Oceanography*, 44 (7), 1854–1872, https://doi.org/ 10.1175/JPO-D-13-0104.1.
- Waterhouse, A. F., and Coauthors, 2022: Global Observations of Rotary-with-Depth Shear Spectra. *Journal of Physical Oceanography*, **52** (12), 3241–3258, https://doi.org/10.1175/JPO-D-22-0015.1.
- Wessel, P., D. T. Sandwell, and S.-S. Kim, 2010: The Global Seamount Census. Oceanography, 23 (1), 24–33.
- Whalen, C. B., J. A. MacKinnon, and L. D. Talley, 2018: Large-scale impacts of the mesoscale environment on mixing from wind-driven internal waves. *Nature Geoscience*, **11** (**11**), 842–847, https://doi.org/ 10.1038/s41561-018-0213-6.
- Winters, K. B., and L. Armi, 2014: Topographic control of stratified flows: upstream jets, blocking and isolating layers. *Journal of Fluid Mechanics*, 753, 80–103, https://doi.org/10.1017/jfm.2014.363.
- Wunsch, C., 1998: The Work Done by the Wind on the Oceanic General Circulation. *Journal of Physical Oceanography*, **28** (11), 2332–2340, https://doi.org/10.1175/1520-0485(1998) 028(2332:TWDBTW)2.0.CO;2.
- Zhang, K., and K. T. F. Chan, 2023: An ERA5 global climatology of tropical cyclone size asymmetry. *International Journal of Climatol*ogy, 43 (2), 950–963, https://doi.org/10.1002/joc.7846.
- Zhao, Z., M. H. Alford, J. B. Girton, L. Rainville, and H. L. Simmons, 2016: Global Observations of Open-Ocean Mode-1 M2 Internal Tides. *Journal of Physical Oceanography*, **46** (6), 1657–1684, https://doi.org/10.1175/JPO-D-15-0105.1.